

A Service of

ZBW

Leibniz-Informationszentrum Wirtschaft Leibniz Information Centre for Economics

Gödl-Hanisch, Isabel; Pandolfo, Jordan

Working Paper Monetary Policy Transmission, Bank Market Power, and Income Source

CESifo Working Paper, No. 11847

Provided in Cooperation with: Ifo Institute – Leibniz Institute for Economic Research at the University of Munich

Suggested Citation: Gödl-Hanisch, Isabel; Pandolfo, Jordan (2025) : Monetary Policy Transmission, Bank Market Power, and Income Source, CESifo Working Paper, No. 11847, CESifo GmbH, Munich

This Version is available at: https://hdl.handle.net/10419/319215

Standard-Nutzungsbedingungen:

Die Dokumente auf EconStor dürfen zu eigenen wissenschaftlichen Zwecken und zum Privatgebrauch gespeichert und kopiert werden.

Sie dürfen die Dokumente nicht für öffentliche oder kommerzielle Zwecke vervielfältigen, öffentlich ausstellen, öffentlich zugänglich machen, vertreiben oder anderweitig nutzen.

Sofern die Verfasser die Dokumente unter Open-Content-Lizenzen (insbesondere CC-Lizenzen) zur Verfügung gestellt haben sollten, gelten abweichend von diesen Nutzungsbedingungen die in der dort genannten Lizenz gewährten Nutzungsrechte.

Terms of use:

Documents in EconStor may be saved and copied for your personal and scholarly purposes.

You are not to copy documents for public or commercial purposes, to exhibit the documents publicly, to make them publicly available on the internet, or to distribute or otherwise use the documents in public.

If the documents have been made available under an Open Content Licence (especially Creative Commons Licences), you may exercise further usage rights as specified in the indicated licence.



WWW.ECONSTOR.EU



Monetary Policy Transmission, Bank Market Power, and Income Source

Isabel Gödl-Hanisch, Jordan Pandolfo



Impressum:

CESifo Working Papers ISSN 2364-1428 (electronic version) Publisher and distributor: Munich Society for the Promotion of Economic Research - CESifo GmbH The international platform of Ludwigs-Maximilians University's Center for Economic Studies and the ifo Institute Poschingerstr. 5, 81679 Munich, Germany Telephone +49 (0)89 2180-2740, Telefax +49 (0)89 2180-17845, email office@cesifo.de Editor: Clemens Fuest https://www.ifo.de/en/cesifo/publications/cesifo-working-papers An electronic version of the paper may be downloaded · from the SSRN website: www.SSRN.com · from the RePEc website: www.RePEc.org

• from the CESifo website: <u>https://www.ifo.de/en/cesifo/publications/cesifo-working-papers</u>

Monetary Policy Transmission, Bank Market Power, and Income Source^{*}

Isabel Gödl-Hanisch[†] and Jordan Pandolfo[‡]

April 22, 2025

Abstract

We provide empirical evidence on banks' market power in financial services and its implications for monetary policy transmission through deposit rates. Banks with market power in financial services charge higher fees for their service and also offer lower deposit rates with less pass-through from monetary policy. We argue that this is the result of product tying: consumers must open a deposit account to access a bank's financial services. We develop and calibrate a quantitative model of the U.S. banking industry where banks generate noninterest income from services in addition to a standard loan-deposit model. Counterfactuals emphasize the importance of non-interest income for credit supply, financial stability, and deposit pricing.

JEL Classification: D43, E44, E52, G21, G51

Keywords: monetary policy, banks, pass-through, market power, product tying

^{*}We are grateful for comments from Viral Acharya, Dean Corbae, Pablo D'Erasmo, Iftekhar Hasan, Ivan Ivanov, Anil Kashyap, Ayşegül Şahin, Jan-Peter Siedlarek (discussant), Laurent Weill, and seminar participants at the Federal Reserve Bank of Kansas City, T2M Amsterdam 2024, CEBRA Annual Meeting 2024, Midwest Macroeconomics Meetings Fall 2024, System Committee on Financial Institutions, Regulation, and Markets Conference 2024, SEA 94th Annual Meeting, and Wisconsin-Queens IO-Finance Reading Group. Views and opinions expressed in this paper reflect those of the authors and do not necessarily reflect those of the Federal Reserve System.

[†]LMU Munich, CEPR, and CESifo. Email: isabel.goedl-hanisch@econ.lmu.de

[‡]Federal Reserve Bank of Kansas City. Corresponding author. Email: jordan.pandolfo@kc.frb.org

1 Introduction

A key question in monetary economics is how monetary policy is transmitted to consumers and firms. While an extensive literature has focused on the pass-through of monetary policy to loan and deposit rates (e.g., Drechsler et al. 2017 and Kashyap and Stein 2000), little focus has been given to the role of non-interest income—an increasingly important source of bank revenue in recent decades. For example, in the United States, non-interest income accounts for approximately 30% of aggregate bank operating income.¹ Much of this income is derived from fees on, or related to, deposit accounts such as fees for managing and transferring funds. Thus, variation in service quality across banks can be an important factor that determines not only non-interest income via fee pricing but deposit pricing, as well.

This paper examines the role of bank non-interest income in determining deposit market power and the pass-through of monetary policy to deposit rates. Using novel branch-level data and local projection methods à la Jordà (2005), we examine how bank rates respond to monetary policy, and how and whether this relationship depends on bank non-interest income. To do this, we interact conventional monetary policy shocks from Jarociński (2024) with a proxy for deposit account fees. While loan rates are unaffected, we find significant results for deposit rates. Specifically, deposit rates at high-non-interest income banks increase by 20 basis points less compared to low-noninterest income banks, given a 100 basis point monetary shock. These results are strongest for accounts associated with financial services, such as checking and savings accounts. Importantly, the non-interest income channel we identify operates independently of traditional drivers of rate pass-through — such as market concentration, capitalization, or size — highlighting the role of financial services as a distinct source of deposit market power.

Beyond our monetary transmission results, we also document that banks with high non-interest income shares exhibit pricing power by setting higher fees and lower deposit rates. We argue that this bank-level correlation is the result of product tying: in order to access the financial services of a particular bank, depositors must first open an account. Thus, a bank with inelastic demand for its services can offer a higher fee *and* a lower deposit rate. This mechanism is consistent with findings in the industrial organization literature whereby non-rate characteristics affect depositor demand (e.g., Egan et al. 2017 and Wang et al. 2022). To illustrate this mechanism, we postulate a simple

¹In the largest euro area countries, the share of non-interest income is similar, or higher, and has been increasing since the Great Financial Crisis (World Bank, 2025).

model where demand for financial services influences deposit pricing through a tying constraint.

We next develop and calibrate a quantitative model of the U.S. banking industry with the novel feature that heterogeneous banks offer services alongside deposits. The key assumption used, based on insights from the illustrative model, is that bank-level demand is correlated across services and deposits.² We use this model to perform counterfactual analyses to understand the role of bank non-interest income for credit supply, financial stability, and deposit pricing.

In the first exercise, we show that the loss of non-interest income leads to a contraction in bank lending and an increase in balance sheet risk. Further, and somewhat paradoxically, low-non-interest income banks are more sensitive to their loss of income. This result occurs because low-non-interest income banks also have less market power, making them more sensitive to the counterfactual. Thus, policies intended to limit bank fee income may have the unintended effect of disproportionately affecting banks with initially low amounts of fee income. This finding contributes to the ongoing policy discourse on capping fees.³

In the second exercise, we demonstrate how non-interest income acts as a risk-free endowment, insulating banks from risks such as interest rate uncertainty. In particular, when interest rate uncertainty increases, all banks contract their lending, but high-non-interest income banks contract lending by less. This result holds in the cross-section of the baseline model but also at the bank level, i.e., in counterfactual exercises when banks lose their non-interest income.

In the third exercise, we examine how changes in the interest rate environment affect banks in the cross-section. Specifically, a decline in the neutral interest rate or slope of the yield curve compresses net interest margins and increases bank risk. While this result is similar to other findings in the literature (e.g., Whited et al. 2021), non-interest income can be a mitigating factor to stabilize profits and bank risk. Overall, the model findings suggest that non-interest income plays a stabilizing role for banks and supports credit supply.

Related Literature. Our research contributes to a vast literature assessing the effect of monetary policy on bank pricing and balance sheet items (e.g., Kashyap and Stein 2000, Bruno and Shin 2015, Altavilla et al. 2020 and Jimenez et al. 2012). We extend this literature by examining the impact of bank non-interest income on the transmission of monetary policy.⁴ More recently,

 $^{^{2}}$ For example, a bank with inelastic demand for financial services is more likely to have inelastic demand for deposits. We assume separate bank-level demand functions for services and deposits to keep the model tractable.

³See, for example, the regulation on non-sufficient funds fees (CFPB, January 2024; Federal Register, January 2024) or credit card late fees (CFPB, March 2024; Federal Register, March 2024).

⁴On the deposit side, stickiness in rates and a sluggish pass-through have been documented, particularly upwards. For example, papers documenting this include Berger and Hannan (1989), Diebold and Sharpe (1990), Neumark

considerable focus has been given to the role of bank and market characteristics, such as size or market concentration, in determining bank pricing power and incomplete pass-through from monetary policy (e.g., Drechsler et al. 2017, d'Avernas et al. 2023, Gödl-Hanisch 2023, Wang et al. 2022, Xiao 2020, Scharfstein and Sunderam 2016, and Narayanan and Ratnadiwakara 2025). We add to this literature by examining how variation in bank non-interest income has meaningful implications for deposit rate pass-through as well as the level of rates and fees associated with deposit accounts.

To perform counterfactual analysis, we construct a dynamic bank model where banks have pricing power. As such, we contribute to a growing class of related models (e.g., Corbae and D'Erasmo 2021, Dempsey 2024, Ulate 2021, Pancost and Robatto 2023, Morelli et al. 2024, and Abadi et al. 2023).⁵ We augment the prototypical loan-deposit bank model by including the provision of financial services and, thus, non-interest income for services.

In large part, we interpret our empirical findings and use of certain model assumptions through the lens of findings from the bank industrial organization literature (e.g. Egan et al. 2017, d'Avernas et al. 2023, Allen et al. 2019, Benetton et al. 2025, and Haddad et al. 2023). In these models, demand for bank deposits or loans is based upon interest rates as well as non-rate characteristics, related to the quality of the bank's product. In this paper, we view variation in the quality of financial services as the ultimate source of bank market power and the reason banks exert pricing power in both deposit rates and the setting of fees.⁶

In addition, our research relates to an empirical literature which examines the relationship between non-interest income and bank risk (e.g., Brunnermeier et al. 2020; DeYoung and Roland 2001; Lepetit et al. 2008; Stiroh 2004, 2006; Stiroh and Rumble 2006). This literature mostly finds a positive relationship and also cites non-interest income as a volatile income source, relative to traditional net interest income. While this may be true for nontraditional activities such as brokerage, insurance, and investment banking, we focus on non-interest income directly related to depositor accounts and services. Thus, in our application, we find a different result: non-interest income is a stabilizing source of income for banks.

The remainder of the paper is structured as follows. Section 2 describes relevant data sources

and Sharpe (1992), Driscoll and Judson (2013), Berlin and Mester (2015), and Yankov (2023).

⁵See also Begenau and Landvoigt 2022, Bianchi and Bigio 2022, Faria-e Castro 2020, Gertler et al. 2020, and Jamilov and Monacelli (2025).

⁶Our simple model also makes use of the literature on product tying (e.g., Tirole 1988, Adams and Yellen 1976, Burstein 1960, Weinberg 1996, Loranth and Morrison 2012).

and definitions. Section 3 presents our main empirical analysis and findings. Section 4 provides an illustrative model to highlight the underlying mechanism which links non-interest income to deposit pricing. Section 5 outlines the main quantitative banking model we use for counterfactual analysis and presents the model results. Section 6 concludes the paper. Details and robustness checks are available in the Online Appendix.

2 Data Description

The primary source for rates and fees is RateWatch, provided by S&P Global Market Intelligence since 2018. RateWatch regularly surveys 76,000 financial institution locations and collects quotes of deposits, mortgages, consumer loan rates, and fees at the branch and product level. The set of rates contains several deposit products: savings, money market, interest checking, CDs of various maturities, and also conditions upon the size of accounts. While this data has been used extensively, we are the first to incorporate RateWatch's collected information on monthly service charges, transaction fees, cash checks, and many more. We further merge the data with the Statistics on Depository Institutions from the Federal Deposit Insurance Corporation (FDIC) and Call Reports to obtain bank characteristics related to income and the balance sheet. We mostly focus on a sample period between 2000 and 2024.

Table 1 provides summary statistics for bank income with a particular focus on non-interest income. On average, non-interest income accounts for 13.2% of banks' total income and exhibits considerable variation in the cross-section of banks. We partition non-interest income into four different categories: *Non-traditional, Asset, Depositor Services* and a residual category called *Other*.⁷ Depositor Services non-interest income refers to fee income generated from financial services provided to owners of deposit accounts and includes payment services, branch services, mobile banking, and penalty fees.⁸ On average, financial services to depositors account for approximately half of all bank non-interest income making it an important source of bank income.

⁷Non-traditional income is generated from activities such as brokerage, insurance, proprietary trading, and investment banking. Asset income is generated from servicing fees, loan origination, and monitoring fees, as well as asset value gains/losses.

⁸Examples of penalty fees include overdraft, minimum balance, and charges for the premature closing of accounts.

Moment	Mean	5p	10p	25p	50p	75p	90p	95p
Int Income to Total Income	86.4	65.3	77.3	84.7	89.5	93.2	96.1	98
Non-Int Income (NII) to Total Income	13.2	2.1	3.8	6.5	10.1	14.7	21.6	32.8
Non-traditional NII to NII	10.5		0	0.4	3.9	12.5	28.3	44.2
Asset NII to NII	8.7	-12.8	-3.7	0	2.2	14.8	34.2	51.0
Other NII to NII	31.0	5.7	9.3	15.1	24.0	40.0	66.3	89.7
Depositor Services NII to NII	49.5	0	3.4	29.8	52.4	70.9	83.0	88.9

TABLE 1: BANK-LEVEL INCOME STATISTICS: 2000-2024

Notes: Cross-section of banks using bank-level time average. Non-traditional NII includes income from brokerage, insurance, proprietary trading, and investment banking. Asset NII includes servicing fees, loan origination fees, monitoring fees, and asset value gains/losses. Depositor Services NII includes fee income generated from financial services provided to owners of deposit accounts and includes payment services, branch services, mobile banking, and penalty fees. Other NII is a residual category. Refer to Appendix B for data definitions. Source: Call Report.

Figure 1 plots the aggregate bank non-interest income share over time and decomposes it into our defined sub-components. The share as well as its sub-components are fairly stable over time. Further, much of the observed variation in the share is driven by either changing interest income or business sources other than depositor services. For example, we find that the cyclical component of aggregate non-interest income from depositor services has a correlation of 0.05 with GDP growth, whereas the other components of aggregate non-interest income have a correlation coefficient of -0.17.⁹ In addition, Figure A.1 shows that bank fees, in levels, are relatively stable over time and invariant to monetary policy. Similarly, at the product level, branch-level monthly fees for savings or money market accounts, as well as fees for transactions such as bill pay, out-of-network and cashier's checks, and fees for incoming and outgoing wire transfers, are acyclical and relatively stable, with some trend growth (see Figure A.2).

⁹We de-trend the aggregate income series and adjust for seasonal factors before computing a Pearson correlation coefficient between non-interest income and GDP growth.

FIGURE 1: DECOMPOSITION OF NON-INTEREST INCOME SHARE OVER TIME



Notes: This table reports non-interest income shares at the aggregate bank level. Source: Call Report.

3 Empirical Evidence

This section uses branch-level and bank-level data to present novel empirical evidence on the relationship between non-interest income, fees, and deposit rates over time. We provide evidence that banks with market power in deposits, as measured by lower deposit rates and lower passthrough of monetary policy, tend to have a higher non-interest income share and more market power in their financial services, as measured by higher fees.

3.1 Empirical Strategy

We use state-dependent local projections to assess the pass-through of monetary policy to deposit rates across banks with different non-interest income shares. We interpret non-interest income shares as a proxy for bank fees. At each horizon h, we estimate the following regression equation for bank i's branch c at time t:

$$r_{t+h,i,c} - r_{t-1,i,c} = \alpha_{i,c}^h + \beta^h s_t + \gamma^h s_t \times NII_{t,i} + \theta^h X_{t,i} + \eta^h Z_t + \epsilon_{t+h,i,c}, \tag{1}$$

where $r_{t+h,i,c} - r_{t-1,i,c}$ represents the cumulative rate change, $\alpha_{i,c}^{h}$ reflects a bank-branch fixed effect, s_{t} stands for a standard monetary policy surprise taken from Jarociński (2024),¹⁰ NII_{t,i} reflects the five-year average non-interest income share,¹¹ and $Z_{t,i}$ is a vector of macroeconomic and financial controls. The vector of macroeconomic and financial controls contains the unemployment rate, industrial production growth rate, CPI inflation, VIX, excess bond premium, and a dummy for the zero lower bound period. As a robustness check, we additionally control for the relevance of other bank-level characteristics. Specifically, we control for the size (log assets) and capitalization (equity ratio) in those specifications with bank-level controls and additionally interact these with the policy surprise.¹² Thus, we can clearly argue that non-interest income is independently important for monetary transmission, *in addition* to size, and capitalization.

The pass-through of monetary policy to deposit rates is defined as the derivative of the change in the deposit rate to the monetary policy surprise and corresponds to the sum of $\beta^h + \gamma^h X_{t,i}$. For the visualization of the results, we focus on two states of $X_{t,i}$: low and high non-interest income shares defined as the 10th and 90th percentile of the non-interest income share distribution.

3.2 Empirical Results

Figure 2 shows that the estimated pass-through of monetary policy shocks to deposit rates is dependent upon a bank's non-interest income share. The figure plots pass-through for four different types of accounts: savings accounts, money market accounts, interest checking accounts, and certificates of deposit. The key finding is that deposit rate pass-through is lower for banks with higher non-interest income shares.¹³

Even though high non-interest income banks exhibit lower pass-through, they experience similar deposit outflows when compared to low non-interest banks (see Figure A.3). Thus, they are able

¹⁰The standard monetary policy surprise captures unexpected movements in the short-end of the yield curve and isolates any information effect. The monetary policy shock is scaled to increase the federal funds rate by one percentage point after 12 months.

¹¹Using the five-year lagged average instead of the contemporaneous share addresses the potential endogeneity concern that banks actively adjust their non-interest income shares to monetary policy.

¹²We use the deviation from the period average to account for secular trends in the size and capitalization.

¹³Refer to Figure A.4 for estimated pass-through across a wider range of deposit accounts.

to maintain similar levels of funding but at a cheaper rate. This allows them to generate a higher net interest margin and a higher return on assets (Figure A.3). To sum up, banks with a larger dependence on financial services fee income (as proxied by non-interest income shares) have greater deposit market power as measured by rate pass-through.





Notes: Impulse responses of deposit rates (savings accounts, money market accounts, interest checking accounts, and certificates of deposit) to a monetary policy shock at both high (blue) and low (green) shares of non-interest income to interest rate income. Horizon is in months. 90% confidence intervals.

The results are most pronounced for savings and interest checking accounts which are exactly the types of accounts most clearly linked to depositor financial services, such as payment services, mobile banking, and branch services. Conversely, certificates of deposit (CDs), which offer a very limited set of services to depositors, show little difference in pass-through.

We also find differences in the level of pass-through *across* different deposit accounts: The pass-

through of monetary policy is the strongest for time deposits, followed by money market accounts, savings accounts, and interest checking accounts, which is in line with findings from Drechsler et al. (2017). While there is almost a complete pass-through to time deposits, particularly for large denominations, savings account rates barely respond. The limited pass-through to savings accounts is due to banks exploiting the inelastic demand of depositors of savings accounts. Moreover, we find that our main empirical finding holds whether we consider only the (i) deposit component or the (ii) non-traditional component of bank non-interest income (Figure A.5).

Robustness. To strengthen the validity of the results and to provide further evidence in support of our channel of financial services relevance for monetary transmission, we perform several crosschecks: (i) using the (log) number of branches and the fee price as a proxy for the quality of financial services instead of the share of non-interest income, (ii) examining the pass-through of internet banks (which offer a very limited range of financial services), and (iii) examining pass-through at the headquarter level.

Consistently, banks with a broader branch network - and therefore more financial services available - pass through monetary policy by less (Figure A.6 and Figure A.7). The difference between the impulse responses of banks with a high vs. low number of branches (10th vs. 90th percentile) is statistically significant but smaller than when considering the share of non-interest income.¹⁴ Alternatively, using the fee price (service charges on deposit accounts relative to deposits) as a proxy confirms that banks with high fees pass through monetary policy less (Figure A.8).¹⁵ Similarly, bricks-and-mortar banks — whether commercial banks, credit unions or savings and loans associations — that offer a wider range of financial services pass through monetary policy to deposit rates by less than internet banks (Figures A.9 - A.11).¹⁶ Our estimates at the headquarter level are very similar to those at the branch level (Figure A.12).

Bank Concentration vs. Non-Interest Income Does the non-interest income channel matter beyond the traditional deposit channel of monetary policy transmission, i.e., after controlling for local market concentration? To address this question, we estimate regressions in the spirit of

¹⁴Of course, the number of branches does not change in the event of a monetary policy shock, but can be regarded as exogenous and therefore serves as a good proxy and instrument.

¹⁵Our preferred proxy is the non-interest income share, as it is less subject to potential measurement error than price data.

¹⁶The sample of internet banks is too small to allow state-dependent local projections. Instead, we compare the pass-through of monetary policy to deposit rates across bank types.

Drechsler et al. (2017), contrasting the role of market concentration and non-interest income in the pass-through to various deposit rates. Specifically, we estimate the change in the deposit rate $\Delta r_{t,i,c}$ to monetary policy conditional on local concentration and share of non-interest income at the branch level:¹⁷

$$\Delta r_{t,i,c} = \alpha_c + \zeta_{c(i)} + \beta \Delta F F_t + \gamma_1 \Delta F F_t \times HHI_{t-1,i,c} + \gamma_2 \Delta F F_t \times NII_{t,i} + \theta X_t + \varepsilon_{t,i,c}, \quad (2)$$

where $\Delta r_{t,i,c}$ denotes the change in the respective deposit rate, ΔFF_t is the change in the federal funds rate, $HHI_{t-1,i,c}$ is the county-level concentration measure, the lagged county-level Herfindahl-Hirschman-Index (HHI), $NII_{t,i}$ is the five-year average of the non-interest income share of bank i, X_t are macroeconomic controls as specified earlier in the local projection equation in (1), and α_c and $\zeta_{c(i)}$ are county and branch fixed effects, respectively.

Table A.1 reports the results across three deposit products. In all panels, the interaction between the federal funds rate change and non-interest income share γ_2 is negative and statistically significant, indicating that banks with a higher reliance on non-interest income are associated with a significantly lower pass-through of rate changes to depositors. This effect is economically sizable and consistent across all product types. A 10 p.p. increase in the non-interest income share is associated with a reduction in pass-through by 0.01-0.02 on impact and accumulates over time (see Figure 2). Notably, the magnitude is independent of estimating the regression jointly with concentration or solely with the non-interest income share.

In contrast, the interaction term for market concentration γ_1 is relatively small in magnitude, as both non-interest income and concentration range between 0 and 1 and have a comparable standard deviation of 0.13 and 0.1, respectively. For savings accounts, it is not statistically significant. For money market and CD accounts, the concentration effect is significant but notably weaker than the non-interest income effect.¹⁸ These findings suggest that non-interest income plays an important role in shaping the deposit rate response to monetary policy, pointing to a different channel of monetary policy transmission that operates independently of market concentration defined at the local level.

 $^{^{17}}$ Adding bank-time fixed effects, as in some of the Drechsler et al. (2017) specifications, would prevent the identification of the marginal effect of the non-interest income share, so we omit them.

¹⁸Overall, the size of the coefficients is in the same order of magnitude as Drechsler et al. (2017), despite using a different sample (1998-2022), monthly data, and a time-varying interaction term. The results are similar using the average county-level HHI.

Other Bank-Level Outcomes. Figure A.3 shows the responses of bank-level deposit flows, lending, assets, net interest margins, and bank profitability (ROA, ROE) to monetary policy for banks with high and low shares of non-interest income. The profitability of banks with high non-interest income rises in response to monetary tightening. Intuitively, if banks with high non-interest income increase deposit spreads in response to monetary tightening, this increases net interest margins and total profitability. There are small effects on deposit and asset flows and on lending: all contract slightly in response to monetary tightening. For the most part, the effects are not statistically different between high and low non-interest income banks. For all banks, the increase in deposit spreads leads to an outflow of deposits and fewer resources for lending. Further, lending rates also increase (Figure A.14), adding to the contraction in lending.

Compared to bank lending, we observe a larger contraction in mortgage-backed securities (MBS) and U.S. government agency obligations in response to a monetary tightening (Figure A.13) which can have implications for mortgage origination activity and pricing, as argued by Drechsler, Savov, Schnabl, and Supera (2024). In our application, banks with high non-interest income reduce MBS by less, offsetting the larger effects on aggregate bank lending. Further, there are some compositional changes and shifts from MBS and government agencies to U.S. treasury securities.

3.3 Characteristics of High Non-Interest Income Banks

Do banks with high non-interest income exhibit pricing power through other means? Table 2 documents average bank characteristics when splitting the sample of banks into categories of *high* and *low* non-interest income share banks (Columns 1 and 2) as well as other partitions. We observe that high share banks charge a higher average fee for depositor financial services *and* a larger deposit spread. Our fee measure is derived from the ratio of deposit services charges to total savings deposits and expressed as a percentage. Thus, a fee of 1.23 for high share banks suggests banks generate 1.23 cents in gross income from depositor financial services for every \$1 in deposits.

Importantly, we also consider other data partitions such as *size* as measured by bank designation in the last three columns. d'Avernas, Eisfeldt, Huang, Stanton, and Wallace (2023) emphasize that bank liquidity services increase in bank size, and this is a source of market power for banks. While our summary statistics are consistent with this finding (i.e., large banks have higher fees, lower deposit spreads, and higher non-interest income), our findings in Section 3.2 hold even after controlling for bank size. Thus, our evidence suggests that bank size alone is not a sufficient indicator of the quality of financial services or non-interest income dependence. This point is further emphasized in the binned scatter Figure A.15, which plots bank size and depositor service non-interest income shares.

	Non-In	t Share	By S	pread	By	Fee	Ву	v Designati	on
Moment	High	Low	Large	Small	Large	Small	GSIB	RBO	CBO
Fee	1.23	0.36	1.10	0.73	1.37	0.36	1.06	0.73	0.99
Deposit Spread	3.45	3.32	4.26	2.39	3.49	3.29	3.54	3.32	3.20
Deposit-to-Asset	75.0	65.3	71.5	71.9	73.9	-68.7	70.5	70.9	76.0
Dividend Yield	0.68	0.58	0.70	0.58	0.72	0.54	0.68	0.68	0.48
ROE	10.0	9.5	10.0	9.7	10.5	9.0	10.3	9.7	9.1
ROA	0.98	1.00	0.98	1.00	1.03	0.93	1.00	1.00	0.95
Total NII Share	31.5	23.4	$-\bar{2}\bar{8}.\bar{2}^{}$	29.2	33.4	$\bar{2}\bar{2}.\bar{5}$	34.7	27.3	16.2
Depositor Services NII Share	9.0	3.1	8.3	5.5	9.4	3.8	6.8	7.9	5.9
TCE Ratio	7.6	9.4	7.8	8.7	7.6	9.1	7.5	8.3	9.9
RW Capital Ratio	13.6	17.7	14.5	15.7	13.7	16.8	14.4	15.1	16.4
Equity Issuance Rate	5.3	12.3	5.0	12.5	4.7	$1\bar{2}.\bar{6}$	11.3	30.8	

TABLE 2: EMPIRICAL MOMENTS: 2000 Q1-2024 Q1

Notes: Unless otherwise specified, all objects are annualized and computed as asset-weighted averages. *Fee Non-Int Share Dep* measures non-interest income using only the deposit service charge line item. The *TCE Ratio* represents the tangible common equity ratio, which accounts for unrealized losses. Source: Call Report.

While high non-interest income banks set higher fees for financial services, it is unclear whether these banks actually have pricing power in financial services. For example, banks may simply spend more for high-quality services and, thus, charge higher fees in a way that is *not* margin-increasing. To address this, we first look at bank-level, time-varying correlations between non-interest income shares, fees and return on equity (see Figure A.16). The results show that banks with high fees and high non-interest income also generate higher return on equity, suggesting they are able to maintain larger profit margins.

We also estimate empirical cost functions to derive fee markups in a method similar to Berger, Klapper, and Turk-Ariss (2017) and Corbae and D'Erasmo (2021). Specifically, for bank loan and deposit quantities $\{q_{it}^{\ell}, q_{it}^{d}\}$ we estimate a translog cost function

$$log(\text{Non-Int Expense}_{it}) = \beta_1 log(q_{it}^{\ell}) + \beta_2 log(q_{it}^{d}) + \beta_3 log(q_{it}^{\ell})^2 + \beta_4 log(q_{it}^{d})^2 +$$
(1)
$$\beta_5 log(q_{it}^{\ell}) log(q_{it}^{d}) + \sum_{j=1}^3 \gamma_j^{\ell} log(w_{ijt}) log(q_{it}^{\ell}) + \sum_{j=1}^3 \gamma_j^{d} log(w_{ijt}) log(q_{it}^{d}) +$$
$$\sum_{j=1}^3 \sum_{k=1}^3 \phi_{jk} log(w_{ijt}) log(w_{ikt}) + \xi_i + \xi_t + \epsilon_{it}$$

where ω_{ijt} is the input price of input j for bank i at time t. We consider three bank input prices:

the wage cost of labor, interest expense of debt, and the cost of physical assets as measured by bank fixed assets. Because the quantity of financial services is unobservable, we make the assumption that services are proportional to observable deposits. While this introduces bias to the estimated level of markups, we care more about the cross-sectional variation in the estimated markup. Then, the estimated marginal cost of providing financial services can be expressed as

$$\widehat{mc}_{it}^{services} = \frac{\text{Non-Int Expense}_{it}}{q_{it}^d} \Big[\hat{\beta}_2 + 2\hat{\beta}_4 log(q_{it}^d) + \hat{\beta}_5 log(q_{it}^\ell) + \sum_{j=1}^3 \hat{\gamma}_j^d log(w_{ijt}) \Big]$$
(2)

and the fee markup is measured as the ratio of net fee income $f_{it} - \hat{m}c_{it}^{services}$ to estimated marginal cost $\hat{m}c_{it}^{services}$. Figure 3 is a binned scatter plot of the relationship between the estimated fee markups and bank deposit spreads, as measured by the difference between the federal funds rate and the rate on savings deposits. It shows a strong, positive relationship: banks with fee pricing power in financial services also have pricing power in deposit rates.¹⁹ This finding is also consistent with the negative branch-level correlation between deposit rates and deposit account fees, documented in Figure A.2.

FIGURE 3: FEE MARKUPS AND DEPOSIT SPREADS BINNED SCATTER



Notes: Markup measures ratio of *fee* minus *marginal cost* to *marginal cost*. Bin averages are based upon bank-year observations. Source: Call Report.

¹⁹It is not an issue that estimated fee markups are less than 1. What matters for the bank is total profitability. For example, the bank may be willing to take financial losses on the services it provides if it generates larger, offsetting profits within the deposit-taking and lending arms of the bank.

Taking Stock. We find that banks with higher non-interest income shares exhibit signs of market power in the level and pass-through of deposit rates, as well as the setting of fees for financial services. Implicitly, banks provide two different products in the form of financial services and deposit accounts which are *tied* together: consumers must open a deposit account to access high quality financial services of the bank.

Strictly as a savings option, insured bank deposits are a relatively homogeneous product across banks (i.e., they offer similar levels of credit risk and liquidity). Thus, in isolation, one would expect deposit products to be priced competitively and uniformly across banks. Instead, there exists significant cross-sectional variation in deposit pricing. We argue this is due to product tying: banks exert deposit market power *through* their financial services market power. Section 4 provides a simple model to illustrate this mechanism.

4 Illustrative Model

In this section, we provide a simple banking model to demonstrate the type of demand behavior that rationalizes the empirical results presented in Section 3. Specifically, we consider a monopolistically competitive banking sector, in which banks provide deposit accounts — a savings vehicle — as well as financial services. We show that when the two products are *tied* (i.e. a consumer must open a deposit account in order to access a bank's services) the bank can exert its services market power in the deposit market and thereby generate lower deposit rates and lower deposit rate pass-through.

Assume consumer preferences are linear in consumption $\{c, c'\}$ and bank services q^n , where q^n is a CES aggregator over different banks indexed by *i* with elasticity ϵ^n

$$c + \beta c' + q^n, \tag{1}$$

and c' is next-period consumption. Consumers choose consumption, deposit savings $\{q_i^d\}$, and services that satisfy a budget constraint

$$c + \sum_{i} \frac{q_i^d}{1 + r_i^d} + \sum_{i} f_i q_i^n = \omega, \qquad (2)$$

where ω is an initial endowment and r_i^d the bank-specific offered net nominal interest rate. We

assume a bank-level tying constraint

$$q_i^n \leqslant \phi q_i^d \tag{3}$$

which requires consumers to hold some quantity of deposits in order to access bank services.²⁰ In the case of a binding constraint, the consumer's first order condition with respect to deposits can be stated as:

$$\beta \left(\frac{q_i^d}{q^d}\right)^{-\frac{1}{\epsilon^d}} + \phi \left(\frac{q_i^d}{\tilde{q}^d}\right)^{-\frac{1}{\epsilon^n}} = \frac{1}{1+r_i^d} + \phi f_i,\tag{4}$$

where $\tilde{q}^d = \left(\sum_i q_i^d \frac{\epsilon^n - 1}{\epsilon^n}\right)^{\frac{\epsilon}{\epsilon^n - 1}}$.²¹ Intuitively, when the tying constraint binds, a consumer may demand additional units of deposits (above their preferred unconstrained amount) in order to obtain more financial services. In this case, the bank is able to exert its service market power on deposit pricing and create an *effective* deposit rate elasticity $\tilde{\epsilon}^d$ which is lower than the *primitive* elasticity parameter ϵ^d .

Special Case. Consider a simplified version of the illustrative model in which $\phi = 1$, $\beta = 1$ and demand is normalized such that $q^d = q^n = 1$. Further, assume that the primitive demand elasticities are the same (i.e. $\epsilon^d = \epsilon^n = \epsilon$). Then, from equation (4), deposit demand can be expressed as $q_i^d = \left(\frac{f_i + \frac{1}{1+r_i^d}}{2}\right)^{-\epsilon}$ and the deposit demand rate elasticity as

$$\frac{\partial q_i^d}{\partial r_i^d} \frac{1+r_i^d}{q_i^d} = \underbrace{\epsilon \frac{\frac{1}{1+r_i^d}}{f_i + \frac{1}{1+r_i^d}}}_{\widetilde{\epsilon}} < \epsilon$$
(5)

such that depositors are relatively more inelastic under the tying regime, i.e. the effective demand elasticity $\tilde{\epsilon}$ is lower than the primitive demand elasticities ϵ .

This simple model illustrates how bank fees and deposit rate pricing are intertwined: banks jointly price their services and deposit accounts as a function of demand parameters for both products. Thus, without conditioning upon the strategic behavior of banks, we would expect to see a negative bank-level correlation between measured demand elasticities $\{\tilde{\epsilon}_i^d, \tilde{\epsilon}_i^n\}$ which generates the negative correlation between fees and deposit rates.

²⁰Unlike the tying literature reviewed in Section 1, we are not explicit about the strategic decision of banks to tie products in order to generate increased market power but, instead, impose this constraint ad hoc. ²¹Our model nests the standard demand of deposits $q_i^d(r_i^d; \epsilon^d)$ when it is assumed that the constraint is not

²¹Our model nests the standard demand of deposits $q_i^d(r_i^d; \epsilon^d)$ when it is assumed that the constraint is not binding $(\phi=0)$: $q_i^d = \beta^{\epsilon^d} (1+r_i^d)^{\epsilon^d} q^d$.

5 Quantitative Model

In this section, we calibrate a quantitative model of the U.S. banking sector and use it to perform counterfactual analysis, testing the implications that bank non-interest income dependence has for credit supply, financial stability, and deposit pricing.

5.1 Bank Problem

Decisions, Constraints, and Technology. There exists I bank types with corresponding probability masses $(p_1, p_2, ..., p_I)$ who monopolistically compete for consumer deposits and financial services. Each bank i has a fixed set of technology and demand parameters. In practice, variation in bank demand could emerge from differences in the quality of financial services, such as the branch network, mobile banking technology, or customer service.

Banks earn profits from a standard deposit-loan balance sheet model as well as the provision of financial services. Each period, banks provide a quantity of financial services q_i^n at a fee price f_i . In addition, banks borrow deposits d_i at a rate r_i^d and originate one-period loans ℓ_i which generate an exogenous, risky return r^{ℓ} . Loan returns can be decomposed into two components

$$r^{\ell} = r(z_{-}, z) + \Delta^{\ell} \tag{7}$$

where $r(z_{-}, z)$ is the monetary policy rate, determined by a simple Taylor rule, and Δ^{ℓ} is a constant spread which is meant to capture a positive-sloping yield curve. The monetary policy rate is determined by the realization of the aggregate shock z in the current period and its value z_{-} in the previous period. We assume the aggregate shock follows an AR(1) process; specifically, $z = \rho_z z_- + \epsilon_z$ where ϵ_z is an iid, mean zero random variable with standard error σ_z .

Each period, banks must satisfy a budget constraint

$$\pi_i + \ell_i + C_i(\pi_i, q_i^n, \ell_i) = n_i + d_i + q_i^n f_i$$
(8)

The right-hand side of equation (8) represents bank funding which consists of networth n_i , deposits d_i , and income from financial services $q_i^n f_i$. Beginning-of-period networth n_i can be thought of as retained earnings or beginning-of-period equity for the bank. Total deposits are determined by the demand function of the bank $d_i = q^d(r_i^d; r, \epsilon_i^d)$, where ϵ_i^d is a bank-specific elasticity parameter and,

similar to Drechsler et al. (2021), demand is sensitive to the level of the monetary policy rate. In addition, demand for financial services is determined via $q_i^n = q^n(f_i; \epsilon_i^n)$ where ϵ_i^n is a bank-specific elasticity parameter. Unlike the illustrative model in Section 4, we keep the model tractable by not explicitly linking demand for services to demand for deposits, but by allowing the elasticity parameters to be correlated at the bank level.

The left-hand side of the budget constraint includes expenses for dividends π_i , loan origination ℓ_i , and operational costs $C_i(\pi_i, q_i^n, \ell_i)$ related to equity issuance, loan origination, and the provision of financial services.²²

The dynamics of the bank problem are captured through the law of motion for next-period networth n':

$$n'_{i} = (1 + r^{\ell})\ell_{i} - (1 + r^{d}_{i})d_{i}$$

$$= \left(1 + r(z, z') + \Delta^{\ell}\right)\ell_{i} - (1 + r^{d}_{i})q^{d}\left(r^{d}_{i}; r(z_{-}, z), \epsilon^{d}_{i}\right)$$
(9)

Note that next-period bank loan returns r^{ℓ} are stochastic and vary with the realized value of the monetary policy rate in the next period such that bank loan rates are adjustable.

Optimization. The bank's objective is to maximize the expected, discounted dividend stream to equity owners subject to the budget constraint, networth law of motion, and the aggregate shock law of motion. Thus, we can express the bank value function as:

$$v(n_{i}, z_{-}, z; i) = \max_{\pi_{i}, r_{i}^{d}, f_{i}, \ell_{i}} \pi_{i} + \beta E[v(n_{i}', z, z'; i)]$$

$$s.t. \quad \pi_{i} + \ell_{i} + C_{i}(\pi_{i}, q_{i}^{n}, \ell_{i}) = n_{i} + q^{d}(r_{i}^{d}; r, \epsilon_{i}^{d}) + f_{i}q^{n}(f_{i}; \epsilon_{i}^{n})$$

$$s.t. \quad n_{i}' = \left(1 + r(z, z') + \Delta^{\ell}\right)\ell_{i} - (1 + r_{i}^{d})q^{d}(r_{i}^{d}; r, \epsilon_{i}^{d})$$

$$s.t. \quad z' = \rho_{z}z + \epsilon_{z}$$

$$(10)$$

Define a bank's state as $\mathbf{s}_i = (n_i, z_-, z; i)$ such that a policy function $y(\mathbf{s}_i)$ represents a currentperiod decision and $y(\mathbf{s}'_i)$ represents a policy function based upon the next period's state. Let $\lambda(\mathbf{s}_i)$ be the shadow multiplier on the bank's current period budget constraint. The bank's first-order

 $^{^{22}}$ Equity or, more broadly, dividend adjustment costs introduce an important financial friction that helps capture the leverage dynamics faced by banks.

conditions with respect to pricing and loan origination can be stated as:

$$[r_i^d]: \quad \frac{\partial d_i}{\partial r_i^d}(\mathbf{s}_i) - E[m(\mathbf{s}_i')] \left(r_i^d(\mathbf{s}_i) \frac{\partial d_i}{\partial r_i^d}(\mathbf{s}_i) + d_i(\mathbf{s}_i) \right) = 0 \tag{11}$$

$$[f_i]: \quad f_i(\mathbf{s}_i)\frac{\partial q_i^n}{\partial f_i}(\mathbf{s}_i) + q_i^n(\mathbf{s}_i) - \frac{\partial C_i}{\partial q_i^n}\frac{\partial q_i^n}{\partial f_i}(\mathbf{s}_i) = 0$$
(12)

$$[\ell_i]: -\left(1 + \frac{\partial C_i}{\partial \ell_i}(\mathbf{s}_i)\right) + E\left(m(\mathbf{s}'_i)\left(1 + r(z, z') + \Delta\right)\right) = 0, \tag{13}$$

where $m(\mathbf{s}'_i)$ is the bank stochastic discount factor, defined as $m(\mathbf{s}'_i) = \beta \frac{E[\lambda(\mathbf{s}'_i)]}{\lambda(\mathbf{s}_i)}$ and $\lambda(\mathbf{s}_i)$ is determined by the dividend equilibrium condition $1 - \lambda(\mathbf{s}_i) \left(1 + \frac{\partial C_i}{\partial \pi_i}(\mathbf{s}_i)\right) = 0.$

For added context in how banks set prices, consider the case in which deposit and financial services demand are characterized by constant price elasticities of (e^d, e^n) where $e^d > 1$ and $e^n > 1$. Further, consider a non-stochastic equilibrium in which $\lambda(\mathbf{s}_i) = \lambda(\mathbf{s}'_i)$ in all states. Then, deposit rates and fees are set as constant markdowns and markups, respectively:

$$[r_i^d]: \quad 1 + r_i^d = \frac{1}{\beta} \frac{e^d}{1 + e^d}$$
(11a)

$$[f_i]: \quad f_i = \frac{e^n}{e^n - 1} \frac{\partial C_i}{\partial f_i} \tag{12a}$$

For deposits, the markdown is with respect to the inverse of the discount factor β^{-1} and in the case in which the monetary policy rate is defined as $1 + r = \beta^{-1}$, banks have a positive deposit spread $s^d = r - r^d = (1 + r)\frac{1}{1+e^d}$ which has imperfect pass-through of monetary policy given $\frac{\partial s^d}{\partial r} = \frac{1}{1+e^d} < 1.$

5.2 Calibration

Functional Forms. For bank demand, we specify financial services demand as a CES demand function with elasticity ϵ_i^n such that $q_i^n = q^n(f_i; \epsilon_i^n) = q^n(f_i^n)^{-\epsilon_i^n}$ which provides a constant price elasticity of $-\epsilon_i^n$. Further, for deposit demand, we specify a logistic functional form

$$q^{d}(r_{i}^{d}; r, \epsilon_{i}^{d}, \xi_{i}^{d}) = \frac{exp(\epsilon_{i}^{d}r_{i}^{d} + \xi_{i}^{d})}{exp(\epsilon_{i}^{d}r) + exp(\epsilon_{i}^{d}r_{i}^{d} + \xi_{i}^{d})}$$
(13)

which is decreasing in the monetary policy rate r, and ξ_i^d is added to represent non-rate characteristics that depositors value. We choose this functional form as it provides a tractable way to account for the monetary policy rate as an outside option for depositors, and it helps generate the pricing dynamics observed in the data. Further, the demand function can be restated in terms of the deposit spread $s_i^d = r - r_i^d$ such that $q^d(r_i^d; r, \epsilon_i^d, \xi_i^d) = \frac{exp(-\epsilon_i^d s_i^d + \xi_i^d)}{1 + exp(-\epsilon_i^d s_i^d + \xi_i^d)}$.²³

In terms of operational costs, we specify the following functional forms:

$$C_i(\pi_i, q_i^n, \ell_i) = \phi_i^{\pi}(\pi_i - \bar{\pi}_i)^2 + q^n(f_i; \epsilon_i^n) mc^f + mc^\ell \ell_i^2,$$
(14)

where banks have a convex cost in adjusting dividends away from their long-run target, a convex cost in loan origination, and linear costs in services.

As noted earlier, the aggregate shock process for loan returns is AR(1) and of form $z' = \rho_z z + \epsilon_z$ where ϵ_z is an iid, mean zero random variable with standard deviation σ^z . Further, monetary policy follows a Taylor rule

$$1 + r' = (1 + r^*) \left(\frac{z'}{z}\right)^{\phi_z} \epsilon^r, \tag{15}$$

where the log of ϵ^r is an iid monetary policy shock with zero mean and standard deviation σ^r and r^* represents the neutral rate of interest in the economy.

External Calibration. Model parameters can be summarized by the bank *i*-specific set

$$\Theta_{i} = \left\{ \underbrace{r^{*}, \phi_{z}, \sigma^{r}}_{\text{monetary policy loan returns}}, \underbrace{\rho_{z}, \sigma^{z}, \Delta^{\ell}}_{\text{bank costs}}, \underbrace{mc^{\ell}, mc^{f}_{i}, \phi^{\pi}_{i}, \bar{\pi}_{i}}_{\text{demand}}, \underbrace{\epsilon^{d}_{i}, \xi^{d}_{i}, \epsilon^{n}_{i}}_{\text{Discount}}, \underbrace{\beta}_{\text{Discount}} \right\}$$
(16)

where most cost and demand parameters are unique to a bank's specific type $i \in I$. Specifically, there are six bank-specific parameters $\{mc_i^f, \phi_i^{\pi}, \bar{\pi}_i, \epsilon_i^d, \xi_i^d, \epsilon_i^n\}$. For the current calibration, we set I = 2 and partition banks according to their dependence on fee income, splitting banks into *high* and *low* non-interest income share buckets according to the median, as reported in Table 2.

The set of model parameters can be partitioned into those which will be externally calibrated and those which will be internally calibrated; i.e., values determined by matching model moments with empirical counterparts. Parameters related to monetary policy implementation $\{r^*, \phi_z, \sigma^r\}$

²³This also allows for a simple representation of the rate elasticity $\frac{\partial q_i^d}{\partial r_i^d} \frac{r_i^d}{q_i^d} = \epsilon_i^d r_i^d (1 - q_i^d)$ and spread elasticity $\frac{\partial q_i^d}{\partial s_i^d} \frac{s_i^d}{q_i^d} = -\epsilon_i^d s_i^d (1 - q_i^d)$.

are externally calibrated based upon common values within the literature.²⁴ Further, we externally estimate the values of $\{\rho_z, \sigma^z\}$ by filtering real GDP and then setting the loan return spread Δ^{ℓ} to achieve an average spread observed in the data. We set the neutral rate of interest in the monetary policy rule to the inverse of the discount factor such that $r^* = \beta^{-1} - 1$. The remaining external parameters are listed in Table 3.

Parameter	Label	Value	Source/Target
β	Discount Factor	0.995	2% annual rate
$ ho_z$	Agg Shock Persistence	0.89	Fernald (2014)
σ_z	Agg Shock Volatility	0.0138	Real GDP Growth (1980s-Present)
ϕ_z	MP Exponential Term	0.9	Literature
Δ^ℓ	Loan Spread	0.016	Corporate Loan Spread BofA

TABLE 3: EXTERNAL CALIBRATION PARAMETERS

Notes: Model parameters are set for a model in quarterly frequency.

Internal Calibration. The remaining set of parameters for the internal calibration are $\{\epsilon_i^d, \xi_i^d, \epsilon_i^n, \phi_i^\pi, mc_i^f, mc^\ell\}$. Given the functional form assumption for financial services demand, the equilibrium condition simplifies to

$$f_i = \frac{\epsilon_i^n}{\epsilon_i^n - 1} m c_i^f \tag{17}$$

such that total services profit enters the period budget constraint as $\pi_i^n = q_i^n(f_i; \epsilon_i^n)(f_i - mc_i^f)$ which is constant and invariant to the bank's state \mathbf{s}_i . In this way, services net income π_i^n acts as a risk-free period endowment. While this pricing condition may seem unrealistic, it is entirely consistent with the observed invariance of bank fees, as shown in Figure A.1, as well as the documented acyclicality of non-interest income from depositor services. In the model, this generates an indeterminacy in the level of $\{mc_i^f, \epsilon_i^n\}$. To manage this, we set the marginal cost terms mc_i^f to the estimated values from the translog cost estimation in Section 3.3 at 1%. We then internally calibrate ϵ_i^n to match the moments related to bank non-interest income shares. As shown in Table 4, we recover $\epsilon_{high}^n = 1.025 < 1.055 = \epsilon_{low}^n$. While the implied markups are large, we care less about their levels and more about the cross-bank differences, mainly, the fact that high non-interest income banks have market power in financial services.

²⁴We set $\sigma^r = 0$ in the baseline model so that the source of variation in the policy rate is the aggregate shock process z.

Parameter	Label	Value	Target	Data	Model
ϵ^d_{high}	Rate Elasticity	1.77	Deposit Spread	3.45	3.47
ϵ_{high}^n	Services Elasticity	1.025	Non-Int Share	32	34
ξ^d_{high}	Deposit Shifter	1.12	Deposit-Asset Ratio	75	75
ϕ^{π}_{high}	Dividend Adjustment	0.27	Dividend Ratio	1.55	1.12
ϵ^d_{low}	Rate Elasticity	1.64	Deposit Spread	3.32	3.33
ϵ_{low}^n	Services Elasticity	1.055	Non-Int Share	23	23
ξ^d_{low}	Deposit Shifter	0.67	Deposit-Asset Ratio	65	62
ϕ^{π}_{low}	Dividend Adjustment	0.42	Dividend Ratio	1.51	1.15
mc^ℓ	Loan Cost	0.014	Return on Equity	10	8.3

TABLE 4: INTERNAL CALIBRATION PARAMETERS

Notes: Model parameters are set for a model in quarterly frequency. Data and Model moments are quoted in annualized terms.

In terms of deposit demand, we set the rate elasticity parameter ϵ_i^d to target the average deposit spread $s_i^d = r - r_i^d$, and set the non-rate demand parameter ξ_i^d to target the average deposit-toasset ratio. The calibrated parameters capture the feature that our empirical evidence in Section 3 supports: high non-interest income banks operate with market power in both financial services and deposits. From model simulations, the average rate elasticity with respect to deposits is 1.065 for high non-interest income banks and 1.099 for low non-interest income banks.

The dividend adjustment parameter ϕ_i^{π} affects the flexibility in which banks can issue equity and, thus, affects the rate at which banks can substitute between debt and equity funding over time. We set this parameter to target the ratio between the 90th and 10th percentile of bank dividends such that banks with low ϕ^{π} have less of a financial friction and larger variations in dividends.

Last, we set the loan marginal cost parameter to match the average return on equity for the entire banking sector. While the parameters for loan returns, monetary policy, and deposit demand determine a bank's net interest margin, variation in mc^{ℓ} affects total profitability as captured by measures such as return on equity (ROE) and return on assets (ROA).

Baseline Model Output and Validation. Figure 4 plots simulated cross-sectional moments from the calibrated quantitative model by bank type. Specifically, we see that high non-interest income banks operate with, on average, lower deposit rates and hence larger deposit spreads (left panel).

Given the strength of their deposit demand, high non-interest income banks also use more deposit funding (center panel) compared to their low non-interest income counterparts.



FIGURE 4: SIMULATED CROSS-SECTIONAL MOMENTS

Notes: The left panel shows the distribution of deposit spreads for high and low non-interest income banks. The right panel shows the distribution of non-interest income shares for high and low non-interest income banks.

Figure 5 plots the simulated relationship between important bank moments and the monetary policy rate. For both types of banks, there is a positive relationship between the monetary policy rate and the deposit spread (left panel): as monetary policy rates increase, banks increase deposit rates to remain competitive with the outside option in $q^d(r_i^d; r, \epsilon_i^d, \xi_i^d)$ but the spread widens. Put differently, deposit rates exhibit incomplete pass-through from changes in the monetary policy rate. The net effect is that deposits flow out of the banking sector (center panel) as the policy rate increases.



FIGURE 5: SIMULATED MONETARY POLICY TRENDS

Notes: The left panel shows the deposit spread for high and low non-interest income banks at different levels of the policy rate. The center panel shows the non-interest income share for high and low non-interest income banks at different levels of the policy rate. The right panel shows bank deposits for high and low non-interest income banks at different levels of the policy rate.

Further, when the policy rate is low, banks operate with smaller deposit spreads and, thus, a lower net interest margin. This makes the traditional loan-deposit business model less profitable and increases the relative importance of fee income. Thus, non-interest income is a larger component of total income when the monetary policy rate is low (right panel).

We also simulate the cumulative impact of a one-time unanticipated 100 basis point monetary policy shock to see how equilibrium objects in the model respond. As shown in Figure 6, deposit rates behave similarly to the estimated effects from Section 3.2. For deposit rates, while the model captures the feature that high non-interest income share banks have lower pass-through, the relative difference between the two bank types is more modest. This is not a significant shortcoming of the model as the empirical estimation is based upon more extreme values (i.e. the 10th and 90th percentile of banks according to non-interest income shares) while our calibration is based upon a median threshold. FIGURE 6: 100 BASIS POINT MP SHOCK (MODEL)



Notes: Lines reflect model-implied responses of deposit rates and deposits to a 100 basis point increase in the policy rate for low and high non-interest income share banks.

5.3 Counterfactual Analysis

In this section, we perform counterfactual exercises to understand the role of fee income for bank credit supply, financial stability, and deposit pricing. In the first counterfactual exercise, we measure the effect on banks of eliminating non-interest income. In the second exercise, we examine how changes in rate uncertainty affect both high and low non-interest income banks. Last, we consider the effect of changing the neutral interest rate and slope of the yield curve. This allows us to examine how policy-invariant non-interest income can affect banks in a high or low interest rate environment.

The Loss of Non-Interest Income. For this counterfactual, we set the services elasticity parameter sufficiently high such that $f_i = mc_i^f$ and banks earn zero profit from their financial services. Table 5 records the outcomes for both bank types in levels while Table 6 reports changes, relative to the baseline model.

Lost non-interest income represents a significant decline in bank funding. Banks can respond by either shrinking the size of their balance sheet, increasing deposit funding, or decreasing dividends. In the counterfactual, banks respond with a combination of these options: total lending declines and deposit funding increases. For the latter, this can be seen through banks offering a lower deposit spread to attract more deposit funding. The net result is a more leveraged balance sheet with more risk, as measured by the decrease in bank z-scores. Thus, the loss of non-interest income is associated with increased bank risk and lower credit supply.

	High NII Banks		Low NII Banks		
Object	Fee Income	No Fee Income	Fee Income	No Fee Income	
Capital Ratio	8.0	5.1	12.4	8.5	
Deposit Spread	3.47	3.39	3.33	3.21	
Return on Equity	8.3	9.4	6.1	6.7	
Z-Score	6.3	4.5	9.4	6.8	
Lending	1.0	0.98	1.0	0.97	
Non-Int Inc Share	34	0	23	0	

TABLE 5: FEE INCOME VERSUS NO FEE INCOME MODEL MOMENTS (LEVELS)

Notes: This table reports the level of model moments, for both the *high non-interest income share* (High NII) and *low non-interest income share* (Low NII) banks when banks do not generate fee income. The *Lending* row is normalized to 1 for the baseline scenario with fee income. Reported capital ratios are adjusted to account for non-deposit debt funding.

In addition, a paradoxical result emerges when examining cross-bank differences. Specifically, low non-interest income banks exhibit more sensitivity to the *no fee income* counterfactual. On the surface, this result seems contradictory: high non-interest income banks are the ones losing a larger share of their total income and should be more affected. While this is true, high non-interest income banks also have more deposit market power and are, thus, better insulated from the *no fee income* scenario.

As a result, while high non-interest income banks decrease total lending by 1.9% in the counterfactual, low non-interest income banks decrease lending by 2.8%. Further, the deterioration in financial stability is more pronounced for low non-interest income banks: their z-scores fall by 2.6, as opposed to 1.9. Recently, a popular policy proposal is to cap or limit bank non-interest income with the intention of protecting consumers from harmful fees. While our model cannot capture the nuance of specific fees or policy applications, it does emphasize that an unintended consequence of such policies may be the disproportionate effect it has on low non-interest income banks.

	High NII Banks		Low NII Banks		
Object	Fee Income	No Fee Income	Fee Income	No Fee Income	
Capital Ratio	_	-2.9	—	-3.9	
Deposit Spread	—	-0.08	—	-0.12	
Return on Equity	-	+1.1	-	+0.62	
Z-Score	-	-1.8	-	-2.6	
Lending	—	-1.9	—	-2.8	
Non-Int Inc Share	_	-34	_	-22	

TABLE 6: FEE INCOME VERSUS NO FEE INCOME MODEL MOMENTS (CHANGES)

Notes: This table reports the change in the level of model moments, for both the *high non-interest income share* (High NII) and *low non-interest income share* (Low NII) banks when banks do not generate fee income. Change in lending is reported as *percentage change* relative to the baseline fee income scenario.

Elevated Uncertainty. In the second counterfactual exercise, we examine how increased rate uncertainty affects bank credit supply. Specifically, we examine how an increase in the volatility of the aggregate shock process σ_z affects the average quantity of bank loans. Higher σ_z translates into higher volatility in the monetary policy rate, which affects both deposit demand and loan returns.

Figure 7 plots the average change in lending, relative to the baseline model, for both bank types. In the counterfactual, all banks respond to higher uncertainty by reducing their lending but low non-interest income banks, again, are more sensitive and decrease lending by more. This result can be explained through understanding bank risk tolerance and their dependence on non-interest income.

Given a convex cost function, bank value functions are concave and, thus, banks exhibit risk aversion. In the model, non-interest income acts as a risk-free endowment which is beneficial to the bank. Given that high non-interest income banks have a larger risk-free endowment, they are less affected by an increase in interest rate risk in the counterfactual. This point is further illustrated when observing the *no fee income* counterfactual scenarios in Figure 7. FIGURE 7: RETURN UNCERTAINTY AND PERCENTAGE CHANGE LENDING, BY BANK TYPE



Notes: This figure shows the change in lending as the rate/return uncertainty σ_z increases for different types of banks in a with and without non-interest income environment. Lending is computed as the simulated within-bank average loan amount and changes are relative to the baseline model.

Collectively, the first two counterfactual exercises emphasize that policy-invariant fee income can be a stabilizing force for banks, supporting lending and financial stability. For other types of non-interest income that exhibit cyclical variation, such as trading revenue or underwriting fees, this may be less the case. For example, see DeYoung and Roland (2001) and Stiroh (2004).

Changing Interest Rate Environment. In the final exercise, we consider the impact of a change in the neutral interest rate r^* and yield curve slope via Δ^{ℓ} .²⁵ Specifically, we examine a *Low* r^* scenario corresponding to a 100 basis point decline in the neutral interest rate, as well as a *Flatter Curve* scenario corresponding to a 50 basis point decline in the loan spread. Table 7 presents the results.

In the Low r^* scenario, deposit spreads decrease by 42 to 47 basis points, leading to an inflow of deposits. Despite the influx of funding, banks become more restrictive on credit issuance and lending contracts. The overall riskiness of banks, as measured by the z-score, increases, and the capital ratio decreases, suggesting greater fragility in the financial system. Similar to the two

²⁵For each change in r^* we also change the discount factor β to be consistent with the inverse of the gross rate.

previous counterfactuals, the results are qualitatively similar across banks but more pronounced for low non-interest income banks.

	High NII Banks		Low	NII Banks
Object	Low r*	Flatter Curve	Low r*	Flatter Curve
Capital Ratio	-9.2	-4.4	-10.8	-3.5
Deposit Spread	42	+0.05	47	+0.06
Z-Score	-5.6	-2.79	-7.1	-2.44
Lending	-1.1	-6.84	-3.5	-6.69
Non-Int Inc Share	+0.3	+1.63	+0.69	+1.27

TABLE 7: LOW VERSUS HIGH INTEREST RATE ENVIRONMENT

Notes: This table reports the change in the level of model moments, for both the high non-interest income share (High NII) and low non-interest income share (Low NII) banks. The Low r^* scenario corresponds to a 100 basis point decrease in the annual neutral interest rate and the *Flatter Curve* scenario corresponds to a 50 basis point drop in bank loan spreads. Change in lending is reported as percentage change relative to the baseline scenario. Reported capital ratios are adjusted to account for non-deposit debt funding.

In the *Flatter Curve* scenario, banks exhibit a much larger balance sheet response, and this is largely due to differences in deposit demand. In both scenarios, banks experience thinner net interest margins, but in the low r^* scenario, the lower monetary policy rate increases deposit demand. This pushes deposits onto a more inelastic part of the demand curve and gives banks increased flexibility to manage their balance sheet. Conversely, in the flatter curve scenario, banks experience only an exogenous drop in loan spreads. The contrast in outcomes between these two scenarios, of course, depends on the extent to which the change in the yield curve is independent of the change in the neutral rate of interest. The main takeaway is that both scenarios— a lower r^* and a flatter curve — increase the volatility of bank aggregates, bank riskiness, and financial instability.

6 Conclusion

We provide novel empirical evidence that the pass-through of monetary policy to bank deposit rates depends upon a bank's reliance on non-interest income. Specifically, banks with higher noninterest income shares have a lower pass-through of monetary policy to deposit rates and charge a higher average level of fees and deposit spreads. We interpret this deposit pricing power as a consequence of market power in financial services, which are tied to deposit accounts. Using a quantitative model of the U.S. banking industry, we further show that non-interest income plays a key role for credit supply, financial stability, and deposit pricing. Future work could collect depositor-level evidence — such as through survey data — to better understand how individuals use and value bundled financial services, and develop more micro-founded models that endogenize the tying constraint in a tractable way.

References

- Joseph Abadi, Markus Brunnermeier, and Yann Koby. The Reversal Interest Rate. American Economic Review, 113(8):2084–2120, 2023.
- W. Adams and J. Yellen. Commodity Bundling and the Burden of Monopoly. Quarterly Journal of Economics, 90, 1976.
- Jason Allen, Robert Clark, and Jean-Francois Houde. Search Frictions and Market Power in Negotiated-Price Markets. *Journal of Political Economy*, 127(4), 2019.
- Carlo Altavilla, Fabio Canova, and Matteo Ciccarelli. Mending the Broken Link: Heterogeneous Bank Lending Rates and Monetary Policy Pass-Through. *Journal of Monetary Economics*, 110: 81–98, 2020.
- Juliane Begenau and Tim Landvoigt. Financial Regulation in a Quantitative Model of the Modern Banking System. *The Review of Economic Studies*, 89(4), 2022.
- Matteo Benetton, Alessandro Gavazza, and Paolo Surico. Mortgage Pricing and Monetary Policy. American Economic Review, 115(3), 2025.
- Allen N. Berger and Timothy H. Hannan. The Price-Concentration Relationship in Banking. The Review of Economics and Statistics, 71(2), 1989.
- Allen N. Berger, Leora F. Klapper, and Rima Turk-Ariss. Bank Competition and Financial Stabiloty. Handbook of Competition in Banking and Financial Services, pages 185–204, 2017.
- Mitchell Berlin and Loretta J. Mester. Deposits and Relationship Lending. The Review of Financial Studies, 12(3):579–607, 2015.
- Javier Bianchi and Saki Bigio. Banks, Liquidity Management and Monetary Policy. *Econometrica*, 90(1), 2022.
- Markus K Brunnermeier, Gang Nathan Dong, and Darius Palia. Banks' Noninterest Income and Systemic Risk. *The Review of Corporate Finance Studies*, 9(2):229–255, 2020.
- Valetina Bruno and Hyun Song Shin. Capital flows and the Risk-Taking Channel of Monetary Policy. Journal of Monetary Economics, pages 119–132, 2015.

- M. Burstein. The Economics of Tie-In Sales. Review of Economics and Statistics, 42, 1960.
- Dean Corbae and Pablo D'Erasmo. Capital Buffers in a Quantitative Model of Banking Industry Dynamics. *Econometrica*, 89(6):2975–3023, 2021.
- Adrien d'Avernas, Andrea L Eisfeldt, Can Huang, Richard Stanton, and Nancy Wallace. The Deposit Business at Large vs. Small Banks. Technical report, National Bureau of Economic Research, 2023.
- Kyle Dempsey. Capital Requirements with Non-Bank Finance. *Review of Economic Studies*, Forthcoming, 2024.
- Robert DeYoung and Karin P Roland. Product Mix and Earnings Volatility at Commercial Banks: Evidence from a Degree of Total Leverage Model. *Journal of Financial Intermediation*, 10(1): 54–84, 2001.
- Francis X. Diebold and Steven A. Sharpe. Post-Deregulation Bank-Deposit-Rate Pricing: The Multivariate Dynamics. Journal of Business and Economics Statistics, 8(3), 1990.
- Itamar Drechsler, Alexi Savov, and Philipp Schnabl. The Deposits Channel of Monetary Policy. The Quarterly Journal of Economics, 132(4):1819–1876, 2017.
- Itamar Drechsler, Alexi Savov, and Philipp Schnabl. Banking on Deposits: Maturity Transformation Without Interest Rate Risk. *The Journal of Finance*, 76(3):1091–1143, 2021.
- Itamar Drechsler, Alexi Savov, Philipp Schnabl, and Dominik Supera. Monetary Policy and the Mortgage Market. Working Paper, presented at Jackson Hole Economic Policy Symposium, 2024.
- John C. Driscoll and Ruth A. Judson. Sticky Deposit Rates. Federal Reserve Board Finance and Economics Discussion Series, 2013.
- Mark Egan, A Hortacsu, and G Matvos. Deposit Competition and Financial Fragility: Evidence from the U.S. Banking Sector. *American Economic Review*, 107(1), 2017.
- Miguel Faria-e Castro. A Quantitative Analysis of the Countercyclical Capital Buffer. FRB St. Louis Working Paper, 2020.

- John G. Fernald. A Quarterly Utilization-Adjusted Series on Total Factor Productivity. *Federal Reserve Bank of San Francisco Working Paper Series*, 2014.
- Mark Gertler, Nobuhiro Kiyotaki, and Anrea Prestipino. A Macroeconomic Model with Financial Panics. *The Review of Economic Studies*, 87(1), 2020.
- Isabel Gödl-Hanisch. Bank Concentration and Monetary Policy Pass-Through. Technical report, CESifo Working Paper, 2023.
- Valentin Haddad, Barney Hartman-Glaser, and Tyler Muir. Bank Fragility when Depositors are the Asset. *Working Paper*, 2023.
- Rustam Jamilov and Tommaso Monacelli. Bewley Banks. *The Review of Economic Studies*, forthcoming, 2025.
- Marek Jarociński. Estimating the Fed's Unconventional Policy Shocks. *Journal of Monetary Economics*, 144:103548, 2024.
- Gabriel Jimenez, Steven Ongena, Jose-Luis Peydro, and Jesus Saurina. Credit Supply and Monetary Policy: Identifying the Bank Balance-Sheet Channel with Loan Applications. American Economic Review, pages 2301–2326, 2012.
- Oscar Jordà. Estimation and Inference of Impulse Responses by Local Projections. American Economic Review, 95(1):161–182, 2005.
- Anil K. Kashyap and Jeremy C. Stein. What do a Million Observations on Banks Say about the Transmission of Monetary Policy? *American Economic Review*, pages 407–428, 2000.
- Laetitia Lepetit, Emmanuelle Nys, Philippe Rous, and Amine Tarazi. Bank Income Structure and Risk: An Empirical Analysis of European Banks. *Journal of Banking & Finance*, 32(8): 1452–1467, 2008.
- Gyongyi Loranth and Alan D. Morrison. Tying in Universal Banks. Review of Finance, 16(2), 2012.
- Juan M. Morelli, Matias Moretti, and Venky Venkateswaran. Geographical Diversification in Banking: A Structural Evaluation. Working Paper, 2024.

- Rajesh Narayanan and Dimuthu Ratnadiwakara. Depositor Characteristics and Deposit Sability. Working Paper, 2025.
- Savid Neumark and Steven A. Sharpe. Market Structure and the Nature of Price Rigidity: Evidence from the Market for Consumer Deposits. *Quarterly Journal of Economics*, 107(2), 1992.
- Aaron Pancost and Roberto Robatto. The Effects of Capital Requirements on Good and Bad Risk-Taking. The Review of Financial Studies, 36(2), 2023.
- David Scharfstein and Adi Sunderam. Market Power in Mortgage Lending and the Transmission of Monetary Policy. Unpublished working paper. Harvard University, 2016.
- Kevin J Stiroh. Diversification In Banking: Is Noninterest Income the Answer? Journal of Money, Credit and Banking, pages 853–882, 2004.
- Kevin J Stiroh. A Portfolio View of Banking with Interest and Noninterest Activities. Journal of Money, Credit and Banking, pages 1351–1361, 2006.
- Kevin J Stiroh and Adrienne Rumble. The Dark Side of Diversification: The Case of US Financial Holding Companies. Journal of Banking & Finance, 30(8):2131–2161, 2006.
- Jean Tirole. The Theory of Industrial Organization. MIT Press, 11th Edition(Chapter 3.3), 1988.
- Mauricio Ulate. Going Negative at the Zero Lower Bound: The Effects of Negative Nominal Interest Rates. *American Economic Review*, 111(1), 2021.
- Yifei Wang, Toni M Whited, Yufeng Wu, and Kairong Xiao. Bank Market Power and Monetary Policy Transmission: Evidence from a Structural Estimation. *The Journal of Finance*, 77(4): 2093–2141, 2022.
- JA Weinberg. Tie-In Sales and Banks. FRB Richmond Fed Economic Quarterly, 82(2), 1996.
- Toni Whited, Yufeng Wu, and Kairong Xiao. Market Power and Bank Risk-Taking in Low Interest-Rate Environments. *Journal of Monetary Economics*, 121:155–174, 2021.
- Kairong Xiao. Monetary Transmission Through Shadow Banks. The Review of Financial Studies, 33:2379–2420, 2020.

Vladimir Yankov. In Search of a Risk-Free Asset: Search Costs and Sticky Deposit Rates. *Journal* of Money, Credit and Banking, 2023.

A Additional Figures and Tables

FIGURE A.1: DEPOSIT RATES AND FEES OVER TIME



Notes: Deposit rate computed as the ratio of total deposit interest expense to total deposits. Fee rate computed as the ratio of service charge income on deposit accounts to total deposits. Numbers are annualized by using comprehensive Q4 expense and income for each fiscal year. Source: FDIC.

FIGURE A.2: BRANCH-LEVEL AVERAGE FEES AND CORRELATION WITH DEPOSIT RATES OVER TIME



Notes: Panel A shows average monthly charges for savings and money market accounts over time. Panel B shows the correlation coefficient of monthly charges and deposit rates for each point in time. Panel C shows average charges for transactions (billpay, out-of-network terminal use, and cashier checks) over time. Panel D shows averages of incoming and outgoing wire transfer charges over time. Converted to yearly frequency (average). Source: RateWatch.

FIGURE A.3: BANK-LEVEL LOCAL PROJECTIONS BY NON-INTEREST INCOME SHARE Panel A: Without controls



Notes: Impulse responses of deposits, loans, assets, net interest margin (NIM), return on assets (ROA) and return on equity (ROE) to a monetary policy shock at both high (blue) and low (green) shares of non-interest income. Panel B includes interactions of size and capitalization and the shock variable as controls. Horizon is in quarters. 90% confidence intervals.

FIGURE A.4: LOCAL PROJECTIONS OF DEPOSIT RATES TO MONETARY SHOCK BY NON-INTEREST INCOME SHARE

Panel A: Without controls



Panel B: With controls



Notes: Impulse responses of different deposit rates to a monetary policy shock at both high (blue) and low (green) shares of non-interest income. Panel B includes interactions of size and capitalization and the shock variable as controls. Horizon is in months. (90% confidence intervals).

(a) Service charges on deposit accounts (b) Additional non-interest income over toover total non-interest income tal non-interest income



Notes: Impulse responses of the MM rate 10K to a monetary policy shock at both high (blue) and low (green) shares of 5-year avg. of service charges on deposit accounts over total non-interest income and 5-year avg. of additional non-interest income over total non-interest income, respectively. Horizon is in months. (90% confidence intervals).





Notes: Impulse responses of deposit rates (savings accounts, money market accounts, checking accounts, and certificates of deposit) to a monetary policy shock at both high (blue) and low (green) number of branches. Panel B includes interactions of size and capitalization and the shock variable as controls. Horizon is in months. 90% confidence intervals. vi

FIGURE A.7: LOG NUMBER OF BRANCHES AS A PROXY FOR PAYMENT SERVICES QUALITY Panel A: Without controls



Notes: Impulse responses of deposit rates (savings accounts, money market accounts, checking accounts, and certificates of deposit) to a monetary policy shock at both high (blue) and low (green) log number of branches. Panel B includes interactions of size and capitalization and the shock variable as controls. Horizon is in months. 90% confidence intervals.



FIGURE A.8: FEE PRICE AS A PROXY FOR PAYMENT SERVICES QUALITY

Notes: Impulse responses of deposit rates (savings accounts, money market accounts, checking accounts, and certificates of deposit) to a monetary policy shock at both high (blue) and low (green) fee prices (calculated as service charges on deposit accounts over deposits: iserchg/dep). Horizon is in months. 90% confidence intervals.



FIGURE A.9: LOCAL PROJECTIONS OF MM 10K RATE TO MONETARY POLICY

FIGURE A.10: LOCAL PROJECTIONS OF SAV 2.5K RATE TO MONETARY POLICY



FIGURE A.11: LOCAL PROJECTIONS OF 12M CD 10K RATE TO MONETARY POLICY





FIGURE A.12: ESTIMATION AT THE HEADQUARTER LEVEL

Notes: Impulse responses of deposit rates (savings accounts, money market accounts, checking accounts, and certificates of deposit) to a monetary policy shock at both high (blue) and low (green) shares of non-interest income. Horizon is in months. 90% confidence intervals.

Panel A: Mon	ey Market (25K)		
$HHI \times dFF$	-0.0719***		-0.0609***
	(-5.36)		(-4.64)
NII $\times \mathrm{dFF}$		-0.111***	-0.112***
		(-7.38)	(-7.39)
r2	0.0942	0.0946	0.0948
Panel B: Certi	ficate of deposit	(10K)	
$HHI \times dFF$	-0.0507***		-0.0435**
	(-3.43)		(-2.81)
	. ,		. ,
$\rm NII \times dFF$		-0.178***	-0.179***
		(-9.90)	(-9.87)
r2	0.229	0.232	0.232
Panel C: Savir	ngs $(2.5K)$		
$HHI \times dFF$	-0.00281		0.000464
	(-0.43)		(0.07)
$\rm NII\timesdFF$		-0.147^{***}	-0.147***
		(-23.17)	(-23.12)
r2	0.0769	0.0816	0.0816

TABLE A.1: RATE PASS-THROUGH, CONCENTRATION AND NON-INTEREST INCOME

Notes: This table reports estimated pass-through of policy rate changes to deposit rate changes — money market (10K), 12-month CD (10K), savings (2.5K) — conditional on the county-level market concentration (HHI) and bank's non-interest income (NII) share, both estimated separately and jointly in one regression. All specifications include branch and county fixed effects, as well as macroeconomic controls. Standard errors clustered at the county level. *, *, and *** indicate significance at the 10%, 5%, and 1% levels.

FIGURE A.13: BANK-LEVEL LOCAL PROJECTIONS BY NON-INTEREST INCOME SHARE: ADD. VARIABLES

Panel A: Without controls



Notes: Impulse responses of mortgage-backed securities, U.S. treasury securities, U.S. government agency obligations in logs and as a ratio over total assets to a monetary policy shock at both high (blue) and low (green) shares of non-interest income. Panel B includes interactions of size and capitalization and the shock variable as controls. Horizon is in quarters. 90% confidence intervals.



FIGURE A.14: LOCAL PROJECTIONS OF LOAN RATES TO MONETARY SHOCK BY NII SHARE

Notes: Impulse responses of the 1-year adjustable mortgage rate to a monetary policy shock at both high (blue) and low (green) shares of non-interest income. Horizon is in months. (90% confidence intervals).

FIGURE A.15: Non-Interest Income Share from Depositor Financial Services and Bank Size



Notes: Dots reflect mean non-interest income share by log assets. Bars reflect the 10th to 90th percentile of non-interest income share by log assets. Source: Call Report.

FIGURE A.16: CORRELATION OF DEPOSIT RATES, FEES, AND NON-INTEREST INCOME AT THE BANK LEVEL



Notes: Deposit rate computed as the ratio of total deposit interest expense to total deposits. Fee rate is computed as the ratio of service charge income on deposit accounts to total deposits. Numbers are annualized by using comprehensive Q4 expenses and income for each fiscal year. For each quarter, compute the cross-sectional correlation and apply a rolling three-year average to smooth out the time series. Source: FDIC.

B Data Definitions and Construction

This section provides an overview of how we construct certain bank statistics, non-interest income shares, and the sub-components as presented in Section 2 or Tables 1 and 2. For most bank data, we rely on merger-adjusted balance sheet and income data from bank Call Reports. Total bank income is computed as the sum of total interest income (RIAD 4107) + total non-interest income (RIAD 4079) + gains (losses) from securities (RIAD 4091).

We partition non-interest income into four categories: *depositor services, non-traditional, asset,* and *other. Depositor services* income consists of service charges on deposit accounts (RIAD 4080), safe deposit box rent (RIAD C015), income and fees from the printing and sale of checks (RIAD C013), income and fees from ATMS (RIAD C016), and bank card and credit card interchange fees (RIAD F555). *Non-traditional* income consists of fees from fiduciary activities (RIAD 4070), trading revenue (RIAD A220), venture capital revenue (RIAD B491), investment banking, advisory, brokerage and underwriting fees and commissions (RIAD B490), insurance fees and commissions (RIAD C386 + RIAD C387), and life insurance activity (RIAD C014). *Asset* non-interest income is composed of net servicing fees (RIAD B492), net securitization income (RIAD B493), net gains/losses on sale of loans and leases (RIAD 5416), net gains/losses on sale of other real estate owned (RIAD 5415), and net gains/losses on sale of other assets (RIAD B496). Last, the *Other* category is a residual term based upon total non-interest income at the bank level.

Bank-level deposit spreads are calculated as the difference between the average federal funds rate (within a quarter) and the deposit rate, measured as the annualized ratio of interest expense to total deposits for savings deposit accounts. The imputed bank-level fee for depositor services is derived from the ratio of depositor service charges (RIAD 4080) to total savings accounts. Bank dividend yields are computed as the annualized ratio of dividends to total assets.